Author's response to reviews

Title: Client perceptions of consent, confidentiality and counseling in provider-initiated and voluntary HIV counseling and testing services in Uganda

Authors:

Rhoda K Wanyenze (rwanyenze@hotmail.com)
David Kyaddondo (kyaddondo@hotmail.com)
John Kinsman (johnkinsman@hotmail.com)
Fredrick Makumbi (fmakumbi@musph.ac.ug)
Robert Colebunders (bcole@itg.be)
Anita Hardon (ahardon@xs4all.nl)

Version: 2 Date: 12 April 2013

Author's response to reviews:

April 12, 2013

Editorial Office
BMC Public Health

Re: Manuscript “Client perceptions of consent, confidentiality and counseling in provider-initiated and voluntary HIV counseling and testing services in Uganda”.

Thank you for the opportunity to revise and resubmit our manuscript for onward transmission to BMC Health Services Research. We have resubmitted the manuscript after making the following proposed changes:

1. We have restructured the abstract according to the BMC guidelines, to include the background, methods, results, and conclusions.
2. We have changed ‘Introduction’ to ‘Background’
3. We have added a separate section on conclusions
4. We have revised the title for the manuscript to better reflect the content of the paper.

We look forward to your feedback.

Sincerely,

Rhoda Wanyenze
Makerere University School of Public Health